Author's response to reviews

Title: Comparing the feasibility, acceptability, clinical- and cost-effectiveness of mental health e-screening to paper-based screening on the detection of depression, anxiety, and psychosocial risk in pregnant women: A study protocol of a randomized, parallel-group, superiority trial

Authors:

Dawn E Kingston (dawn.kingston@ualberta.ca)
Sheila McDonald (sheilaw.mcdonald@albertahealthservices.ca)
Anne Biringer (abiringer@mtsinai.on.ca)
Marie-Paule Austin (m.austin@unsw.edu.au)
Kathy Hegadoren (kmh3@ualberta.ca)
Sarah McDonald (mcdonalds@mcmaster.ca)
Rebecca Giallo (rgiallo@parentingrc.org.au)
Arto Ohinmaa (arto.ohinmaa@ualberta.ca)
Gerri Lasiuk (glasiuk@ualberta.ca)
Glenda MacQueen (gmmacque@ucalgary.ca)
Wendy Sword (sword@mcmaster.ca)
Marie Lane-Smith (mlanesmi@ualberta.ca)
Sander Verldhuyzen van Zanten (vanzanten@ualberta.ca)

Version: 5
Date: 27 November 2013

Author's response to reviews:

attached